Author's response to reviews

Title: Seasonal variation in objectively measured physical activity, sedentary time, cardio-respiratory fitness and sleep duration among 8-11 year-old Danish children: A cross-sectional study

Authors:

Mads F Hjorth (madsfiil@life.ku.dk)
Jean-Philippe Chaput (jpchaput@cheo.on.ca)
Kim Michaelsen (kfm@life.ku.dk)
Arne Astrup (ast@life.ku.dk)
Inge Tetens (intet@food.dtu.dk)
Anders Sjödin (amsj@life.ku.dk)

Version: 2 Date: 8 February 2013

Author's response to reviews:

Dear Editor,
I added some minor corrections
Kind regards
Mads